Author's response to reviews

Title: Barriers to sight impairment certification in the UK: The example of a population with diabetes in East London

Authors:

Rabia Bourkiza (rbourkiza@hotmail.com)
Mala Subash (mala.subash@moorfields.nhs.uk)
Dania Qatarneh (dania.qatarneh@moorfields.nhs.uk)
Joanna Dacosta (joanna.dacosta@moorfields.nhs.uk)
Catey Bunce (catey.bunce@moorfields.nhs.uk)
Tunde Peto (tunde.peto@moorfields.nhs.uk)

Version: 4
Date: 31 July 2014

Author's response to reviews:

Changes made according to the editorial request:

1. Email addresses of all co-authors are included in the title page.
2. Abstract: replaced the 'purpose' section with a 'background' section.
3. Conclusions section has been included.
4. Page 11: changed to "competing interests".
5. Table 1 is included in the manuscript after reference section
6. Additional files removed